Satellite data reveal earlier and stronger phytoplankton blooms over fronts in the Gulf Stream region

Clément Haëck¹, Marina Lévy¹, Inès Mangolte¹, and Laurent Bopp² ¹LOCEAN-IPSL, Sorbonne Université, CNRS, IRD, MNHN, Paris, France ²LMD-IPSL, École Normale Supérieure / Université PSL, CNRS, École Polytechnique, Paris, France

Correspondence: M. Lévy

Abstract. Fronts affect phytoplankton growth and phenology by locally reducing stratification and increasing nutrient supplies. Biomass peaks at fronts have been observed in-situ and linked to local nutrient upwelling and/or lateral transport, while reduced stratification over fronts has been shown to induce earlier blooms in numerical models. Satellite imagery offers the opportunity

- 5 to quantify these induced changes in phytoplankton over a large number of fronts and at synoptic scales. Here we used twenty years of Sea Surface Temperature (SST) and Chlorophyll-*a* (Chl-*a*) satellite data in a large region surrounding the Gulf Stream to quantify the impact of fronts on surface Chl-*a* (used as a proxy for phytoplankton) in three contrasting bioregions, from oligotrophic to blooming ones, and throughout the year. We computed an Heterogeneity Index (HI) from SST to detect fronts, and used it to sort fronts into weak and strong ones based on HI thresholds. We observed that the location of strong fronts
- 10 corresponded to the persistent western boundary current fronts, and weak fronts to more ephemeral submesoscale fronts. We compared Chl-*a* distributions over strong fronts, over weak fronts and outside of fronts in the three bioregions. We assessed three metrics, the Chl-*a* excess over fronts at the local scale of fronts, the surplus in Chl-*a* induced at the bioregional scale, and the lag in spring bloom onset over fronts. We found that weak fronts are associated with a local Chl-*a* excess weaker than strong fronts, but because they are also more frequent they contribute equally to the regional Chl-*a* surplus. We also found that
- 15 the local excess of Chl-*a* was two to three times larger in the bioregion with a spring bloom than in the oligotrophic bioregion, which can be partly explained by the transport of nutrients by the Gulf Stream. We found strong seasonal variations in the amplitude of the Chl-*a* excess over fronts, and we show periods of Chl-*a* deficit over fronts north of 45°N that we attribute to subduction. Finally we provide observational evidence that blooms start earlier over fronts, by one to two weeks. Our results suggest that the spectacular impact of fronts at the local scale of fronts (up to + 60%) is more limited when considered at the
- 20 regional scale of bioregions (less than +5 %), but may nevertheless have implications for the region's overall ecosystem.

Plain Language Summary

Phytoplankton vary in abundance in the ocean over large regions, and with the seasons, but also because of small-scale heterogeneities in surface temperature, called fronts. Here, using satellite imagery, we found that fronts enhance phytoplankton much more where it is already growing well, but despite large local increases the enhancement for the region is modest (5%).

25 We also found that blooms start by one to two weeks earlier over fronts. These effects may have implications for ecosystems.

1 Introduction

Phytoplankton form the basis of marine food webs and are key players in the ocean carbon cycle. The transport of limiting nutrients to the sunlit euphotic layer by advective and convective processes, and the amount of light received by the cells — which is closely related to the stratification of the water column — are two important factors that control their growth. As

- 30 there are marked contrasts in nutrient and light availability in the ocean, it follows that the global ocean can be divided into different regional biomes (or bioregions), characterized by different phytoplankton abundances and seasonality (Longhurst, 2007; Vichi et al., 2011; Bock et al., 2022). The contrasts between biomes are largely explained by consistent physical forcings and environmental conditions (such as nutrient sources), operating at the biome scale, which determine how the two main controlling factors, nutrient and light, limit growth. For example, subtropical gyres are areas where negative wind-stress curl
- 35 induces a deepening of the thermocline and nutricline, resulting in oligotrophic biomes where productivity is relatively constant and low throughout the year. At higher latitudes, where the wind-stress curl is positive, the nutricline is shallower; the strong seasonality of the vertical mixing will induce a multi-stage operation, with a time of reduced productivity and convective nutrient supply in winter when the mixing is strong, and a bloom in spring when the stratification sets in (Wilson, 2005; Williams and Follows, 2011). In addition to these large-scale patterns, there has been considerable evidence over past years
- 40 that the nutrient and light environments are modified at ocean fronts, with consequences on phytoplankton (see reviews by Lévy et al., 2012; Mahadevan, 2016; Lévy et al., 2018).

Ocean fronts are narrow zones where horizontal gradients in water properties (temperature, salinity, nutrients, etc.) are significant (Belkin et al., 2009), and are sometimes described as discontinuities because of their abrupt nature (Mauzole, 2022). Lévy et al. (2018) distinguished two types of fronts. Persistent fronts like those associated with the Gulf Stream and

- 45 Kuroshio are locked in place by the coastal boundary and large-scale atmospheric forcing. Their forcing is directly balanced by submesoscale symmetric instability, which takes energy mostly from the kinetic energy of the jet, and baroclinic instability, which converts the potential energy of the sloping density surfaces into large meanders and eddies. Their ephemeral cousins are continuously forming, moving, and dissipating at the ocean surface. They are being strained by mesoscale eddies, which intensifies their geostrophic along-front currents, and feeds fluid instabilities generating submesoscale vortices and filaments.
- 50 Importantly, a front will generate a cross-frontal ageostrophic secondary circulation an overturning circulation directed in the direction of trying to flatten the density surfaces in the front, with upwelling on the warm side of the front, and downwelling on the cold side (Thomas et al., 2008; McWilliams, 2016; Mahadevan et al., 2020). The highly energetic persistent fronts are characterized by a deep vertical velocity structure that reaches the thermocline, while ephemeral fronts have associated cross-frontal secondary circulations that are generally confined to the vertically well-mixed upper layer of the ocean. Frontal
- 55 dynamics also involves increased stratification, reduction of mixed-layer depth and suppression of vertical mixing at the front (Thomas and Ferrari, 2008).

Fronts may affect phytoplankton in various ways. The upward branch of the secondary circulation may enhance phytoplankton growth by transporting nutrients into the euphotic zone (Johnson et al., 2010; Wilson, 2021). The downward branch may subduct biomass and excess nutrients into the subsurface (Calil et al., 2011; Omand et al., 2015; Hauschildt et al., 2021). Per-

- 60 sistent fronts may act as conduct for nutrients over large distances, known as nutrient streams (Pelegrí et al., 1996; Williams et al., 2011; Long et al., 2022). Advective transport along fronts may transport nutrients to the shallow flanks of subtropical gyres (Letscher et al., 2016; Gupta et al., 2022). Finally, in highly seasonal regimes where productivity is slowed in winter due to deep mixing, increased restratification over fronts may promote localized phytoplankton blooms before the large-scale outburst associated with seasonal stratification (Mahadevan et al., 2012). These effects all together are responsible for local
- 65 anomalies in the distribution of phytoplankton over fronts. Thus fronts are not only a physical boundary but also constitute specific habitats for phytoplankton (Mangolte et al., 2023).

Despite numerous local observations and a strong theoretical basis for the physical processes affecting phytoplankton growth over fronts (e.g. recent studies by Marrec et al., 2018; Little et al., 2018; Verneil et al., 2019; Ruiz et al., 2019; Uchida et al., 2020; Kessouri et al., 2020; Tzortzis et al., 2021), their integrated contribution at the scale of regional biomes is still largely

- 70 unknown. Ephemeral fronts move and dissipate continuously on time scales of days to weeks, and are thus particularly difficult to sample. This limitation is reinforced by the fact that only a limited number of fronts can be observed with in situ field observations. Thus satellite-derived estimates of Chlorophyll-*a* (hereafter Chl-*a*) although limited to the surface of the ocean and an imperfect proxy for phytoplankton biomass, are the only data that allow to track the impact of fronts synoptically over large areas. A first attempt to assess the contribution of small scale physical processes to regional satellite Chl-*a* budgets was
- 75 based on a geostatistical analysis derived from data at 9km resolution (Doney et al., 2003), extended later in Glover et al. (2018), with which they examined the change in spatial variance with distance. This methodology was too coarse to reveal the impact of frontal processes, but confirmed the important role of mesoscale eddies in stirring large scale gradients of phytoplankton abundance. The role of fronts has been assessed with three different methods. Guo et al. (2019) combined ocean color data with altimetry and drifting floats, and estimated that, over subtropical gyres of the global ocean, the respective contributions
- of mesoscale dynamics and submesoscale frontal dynamics to high Chl-*a* anomalies were comparable in magnitude. Keerthi et al. (2022) proposed an approach based on deconvolution of local Chl-*a* time series into different timescales; they observed that sub-seasonal time scales contributed roughly 30% of the total satellite Chl-*a* variance and were associated with small (< 100km) spatial scales which included both the mesoscale and the submesoscale. Finally, the most quantitative approach, and the only one directly related with co-localization with fronts, was proposed by Liu and Levine (2016), which they applied to
- 85 the North Pacific Subtropical Gyre. They detected sea-surface temperature (SST) fronts by computing an index that measures the local heterogeneity of the SST field from satellite data. This allowed them to compare satellite Chl-*a* values over areas impacted by fronts (characterized by a large value of the heterogeneity index) with values over areas that were not impacted. They found that the increase in Chl-*a* over the fronts was negligible in summer but reached almost 40% in winter.

Here we build on this last approach, and quantify the surplus Chl-*a* induced by fronts at the scale of biomes. The excess
90 Chl-*a* over fronts depends on how efficient fronts are at supplying nutrients, which itself depends on how deep the fronts reach into the nutricline, on the seasonality of this vertical supply, and also on the presence of nutrient sources that are advected horizontally along the front. The impact of fronts is also expected to differ between biomes, with submesoscale advection of nutrients likely to be more important in oligotrophic biomes where other nutrient supply routes are scarce, and submesoscale restratification in blooming biomes. Finally, the regional surplus Chl-*a* induced by fronts at the scale of biomes will depend

95 on the spatio-temporal footprint of fronts, which also varies seasonally (Callies et al., 2015) and regionally (Mauzole, 2022). Thus, our intention is to explore and quantify how the contribution of fronts to biome-scale Chl-*a* varies in three contrasted biomes, ranging from subtropical to subpolar, varies along the year, and varies with the occurrence and strength of fronts.

We focus our analysis on the North-Atlantic region surrounding the Gulf Stream, where multiple biomes and fronts of different strengths are found in a limited geographical area (Bock et al., 2022) with strong seasonality. In the south, our

- 100 study area encompasses part of the North Atlantic subtropical gyre, characterized by an oligotrophic regime, year-long low productivity. In the north, north of the Gulf Stream jet, is a more productive subpolar regime characterized by a recurrent spring bloom. In between, there is a moderately productive regime, with maximum productivity in winter. Another feature that makes this study area particularly relevant is that it has two strong persistent fronts, the Gulf Stream and the shelf-break front, which are both associated with strong and deep-reaching vertical circulations (Flagg et al., 2006; Liao et al., 2022), with 105 the Gulf Stream being a recognized horizontal nutrient pathway toward the North Atlantic (Pelegrí et al., 1996; Williams and
- Follows, 2011). But there are also plethora of ephemeral fronts continuously forming at more random locations (Drushka et al., 2019; Sanchez-Rios et al., 2020).

2 Methods and data

We use satellite data of Chl-*a* and SST and extend on the approach of Liu and Levine (2016) to distinguish between persistent
and ephemeral fronts. We evaluate the impact of both types of fronts on Chl-*a* on the basis of three indicators, the excess (or deficit) Chl-*a* over fronts at the local scale of the front, the surplus Chl-*a* attributable to fronts at the scale of regional biomes, and the change in the timing of the Chl-*a* spring bloom over fronts.

2.1 Data

Our approach combines daily satellite SST data, which are used to detect fronts and sort them by their strength, with daily satellite surface Chl-*a*, from which we derive anomalies over fronts. For Chl-*a*, we used the L3 product distributed by ACRI-ST over the period 2000–2020, generated by Copernicus-GlobColour, constructed with data from different sensors (SeaWIFS, MODIS Aqua & Terra, MERIS, VIIRS-SNPP & JPSS1, OLCI-S3A & S3B) merged and reprocessed, available daily at 4km resolution (Garnesson et al., 2019, 2021).

- For SST, we used the European Space Agency Sea Surface Temperature Climate Change Initiative analysis product version 2.1 (Merchant et al., 2019; Good et al., 2020a, b), also available daily at 4km resolution over the period 2000–2020. This product combines data from all available infrared sensors ((A)ATSR, SLSTR, and AVHRR sensors), ensuring good resolution where data are available, unlike other SST products which also include microwave and in-situ measurements, resulting in considerable smoothing of the SST field. Where SST data is not available, spatial interpolation is performed to obtain a cloud free product which, at the cost of resolution on finer features, provides complete synoptic coverage of our large study
- 125 area. This interpolation tends to underestimate the detection of fronts, as the SST field is smoother over cloud-covered areas (Merchant et al., 2019). However, the combination of several sensors allows to reduce these areas to a minimum. Furthermore,

we have only considered cloud-free pixels for our analysis, which ensures that cloudy areas are not taken into account in our quantification.

2.2 Delimitation of biomes

140

130 Our region of interest is the North Atlantic from 15°N to 55°N, and from 40°W to the North American shelf break (Fig. 1). We focus on the open ocean and exclude the continental shelf, thus all pixels where water depth is less than 1500m (red isobath in Fig. 1) are masked. The region is characterized by the presence of a large-scale north-south gradient in Chl-*a* and is sorted into three open-ocean biomes.

The oligotrophic Permanent Subtropical Biome (PSB) — also known as the subtropical gyre permanently stratified biome, 135 (Sarmiento et al., 2004) — to the south of our study area is characterized by warm waters and low Chl-*a* (Fig. 1). There is no clear physical boundary to its northern limit so we have chosen the latitudinal limit of 32°N to delineate it, which roughly corresponds to the $0.1mgm^{-3}$ Chl-*a* iso-contour in annual mean Chl-*a*. There is no persistent front in the PSB.

North of 32° N, the Seasonal Subtropical Biome (SSB) — also known as the subtropical gyre seasonally stratified biome (Sarmiento et al., 2004) and as the permanent deep Chl-*a* maximum biome (Bock et al., 2022) — is also mainly oligotrophic, with intermediate levels of Chl-*a* and temperature, and characterized by slightly increased productivity in winter. It is bounded

- to the north by the meanders of the Gulf Stream jet. When the Gulf Stream enters the SSB, it conveys warm, nutrient rich, salty waters poleward along the Florida coast up to Cape Hatteras (35°N), where it separates from the continental shelf and meanders essentially zonally. The north wall of the Gulf Stream, so called because of its steep temperature gradient, marks the sharp, sinuous and unsteady northern limit of the SSB.
- To the north of the Gulf Stream is the Slope sea which extends to the shelf-break, with colder and fresher waters (Linder and Gawarkiewicz, 1998). Aligned with the shelf break, a persistent front with an intensified surface jet separates the shelf waters (excluded from this study) from the slope sea. This highly productive sub-Polar Biome (PB) also known as subpolar waters (Sarmiento et al., 2004) and as high-chlorophyll-bloom (Bock et al., 2022) is characterized by a strong spring bloom whose onset is tied to the spring stratification of the mixed-layer. The shelf-break front is thus comprised within the PB.
- The position of the north wall of the Gulf Stream, that delimits the SSB and the PB (black meandering contour in Fig. 1ab), is determined at each daily time step by thresholding the daily SST map. The daily threshold values (black vertical line in Fig. 1c) are determined from the daily SST distributions above 32°N (blue line, continued by the yellow line in Fig. 1c). Indeed, the Gulf Stream is easily identifiable in this distribution as it is manifested by a temperature peak (yellow line in Fig. 1c). To detect the start of this peak, which marks our boundary, we fit the peak with a Gaussian, and define the threshold temperature
- 155 as the mean minus twice the standard deviation. Then, the threshold temperature time series is median filtered over an 8-day window to eliminate spurious detection anomalies. This separation method ensures quasi-unimodal Chl-*a* distributions within each biome (Fig. 1d).



Figure 1. Delimitation of the three biomes in the open ocean Gulf Stream extension region: the Permanent Subtropical Biome (PSB, south of the dashed line at 32°N), the Seasonal Subtropical Biome (SSB, between 32°N and the meandering Gulf Stream northern wall on that day marked with the black contour), and the sub-Polar Biome (PB, north of the Gulf Stream northern wall). (a) SST and (b) Chl-*a* snapshots on the 22 April 2007 (with data masked by clouds in white). The red line follows the 1500m isobath. Data on the continental shelf (< 1500m) is not considered here and have been masked. (c) SST and (d) Chl-*a* distribution within each biome for the same day (PB: blue, SSB: yellow, PSB: red). The black line in (c) shows the SST threshold value detected to delimit the Gulf Stream northern wall (see methods section).

2.3 Front detection

Our front detection method aims to sort the domain into pixels that are located (or not) over fronts, at each daily time step.

- We follow the approach of Liu and Levine (2016) that builds upon the well established Cayula-Cornillon algorithm (Cayula and Cornillon, 1992; Belkin and O'Reilly, 2009), and identify pixels as belonging to fronts when the region in the vicinity of the pixel is characterized by high SST heterogeneity. More precisely, on each pixel and for each time step, we compute an Heterogeneity Index (HI) which quantifies the heterogeneity of the SST in a square window comprising a few pixels (7 \times 7) centered on the pixel of interest. Thus HI measures the SST heterogeneity at spatial scales less than 30km. This spatial scale
- 165 captures the heterogeneity associated with sub-mesoscale fronts and persistent fronts (which are wider than sub-mesoscale fronts), with the edges of mesoscale eddies (which are also considered as fronts), but not with the centre of eddies as their diameter is generally >100 km (see for instance Contreras et al. (2023) which nicely shows the scales of these different features in this region). The HI is defined as the weighted sum of the skewness γ , standard deviation σ , and bimodality *B* of SST within the window (HI = $a (b\gamma + c\sigma + dB)$), with a, b, c, d constant normalization coefficients).
- We computed the bimodality as the L2 norm of the difference between the SST histogram (with bins of 0.1°C) and a Gaussian fit of the histogram. We chose this method because it was simpler to implement and more robust than that of Liu and Levine (2016) which computed the normalized absolute difference between a polynomial fit of order 5 of the SST histogram and a Gaussian fit of the histogram, the polynomial fit often being poorly constrained. We normalized each component by its variance (b, c and d being defined as the inverse of the standard deviation of each component computed over one year), as opposed to
- 175 Liu and Levine (2016) who normalized the coefficients by their annual maximum. This avoided putting too much weight on extrema. Finally we normalized HI (coefficient a) such that 95% of values are below an arbitrary value of 9.5. For simplicity, the normalization coefficients were computed for year 2007 and used over the entire time series. An example of the resulting HI over a single front is given on Fig. 2; elevated HI values are located along the curved SST gradient to the north east of the window. We performed sensitivity tests on the different parameters used to compute HI, namely the number of pixels in the
- 180 rolling window and the choice of normalization coefficients. Our results are weakly sensitive to the choice of these parameters (supplementary Fig. A1).

At each time step, we sorted the pixels into those belonging to strong fronts (defined as HI > 10), those belonging to weak fronts (defined as 5 < HI < 10), and those that do not belong to fronts (when HI < 5, and called them "background" in the following). Sorting fronts by range of HI enabled us to roughly separate the persistent fronts, which are associated with the strongest SST gradients (strong fronts), from the ephemeral fronts, associated with weaker gradients (weak fronts). The choice

185 strongest SST gradients (strong fronts), from the ephemeral fronts, associated with weaker gradients (wea of the two HI threshold is somehow arbitrary, but it is supported by the HI distributions presented below.

190

2.4 Quantification of the impact of fronts on Chl-a

In order to quantify how Chl-*a* is affected by fronts at the local scale of fronts, we compared the distributions of Chl-*a* over fronts and the distributions of Chl-*a* outside of fronts. Distributions were computed over 8-day windows to limit the influence of particularly cloudy days. They were computed within each biome, and over latitudinal bands of width 5°, to minimize the

7



Figure 2. The SST, Chl-*a*, and heterogeneity index (HI) of a front on the 7 July 2007. The plain and dashed contours correspond to HI values of 5 and 10. This front is categorized as weak. Chl-*a* are elevated inside the front.

influence of the large-scale north south gradient in Chl-*a*. Distributions were compared in terms of their median value, but using mean values yielded similar results (see suppl. Fig. A2 for seasonal distributions and their medians over and outside of fronts within the three biomes). We defined and computed the local excess Chl-*a*, *E* (expressed in per cent), as the median value over fronts minus the median value in the background, divided by the median value in the background, for each distribution. We repeated this for weak and strong fronts.

195

200

To quantify the large-scale impact of fronts on Chl-*a* at the scale of biomes, we computed the biome surplus of Chl-*a S* which we defined and computed as the relative difference (expressed in per cent) between the mean Chl-*a* over the entire biome (MT), and the mean Chl-*a* over the background (MB), S = (MT - MB)/MB. Thus the surplus *S* measures the extra quantity of Chl-*a* at the scale of the biome (MT - MB), relative to what would be the situation in the absence of fronts (MB), and thus accounts for the local excess over fronts (E), but also for the proportion of fronts in a given biome. To better understand the meaning of the surplus, let us consider the simplified case where Chl-*a* is homogeneously doubled over fronts compared to the background value, i.e. when E = 100%; in that case, the surplus Chl-*a* is 50% if there are 50% of fronts, and is 1% if there are 1% of fronts. Note that the computation of the local excess *E* is based on median values because it relies on the comparison of distributions, while for the computation of the biome surplus *S*, we used mean values in order to be conservative.

- Finally, the subpolar biome is characterized by a spring bloom, of which we measured the onset date both over fronts and in the background, for each year. We defined the lag in bloom onset L as the bloom onset day over fronts minus the bloom onset day in the background. Because the spring bloom onsets propagate from south to north, bloom onset days were inferred over latitudinal bands of limited width (5°). We pooled apart front and non front pixels and computed the onset dates and their uncertainty based on the time series of the fronts and non-front Chl-*a* median value. We first filtered the Chl-*a* median
- 210 time series with a low-pass Butterworth filter of order 2 and cutting frequency $1/20 \text{ days}^{-1}$. The filtered time-series displayed strong variations in their phenology from year to year, but a bloom was always discernible. We considered data from February to July, which allowed us to isolate the spring bloom and exclude the autumn bloom. We detected the maximum value of Chl-*a* in this time window, and defined the bloom onset as the time of maximum Chl-*a* derivative prior to the time of maximum



Figure 3. Normalized distribution of the Heterogeneity Index (HI, red dashed line) within each biome, and distribution of Chl-*a* as a function of HI (representing front strength), over the full period 2000–2020. Shown are the median value of the Chl-*a* distributions (solid black line), and 1st and 3rd quartiles (dashed lines). Note that 0.5% of pixels have outstanding large HI values and are not included here.

Chl-a. We defined the uncertainty in bloom onset date as the standard deviation of all days for which the Chl-a derivative is
above 90% of its maximum value. We defined the uncertainty in lag L as the square root of the sum of squared uncertainties in fronts and background. Finally, we estimated the mean lag (over the 20 years of data and for each latitudinal band) as the weighted averaged lag, with the weights equal to the inverse of the uncertainties; and we estimated the mean lag uncertainty as the weighted standard deviation of all lag values around the mean lag, with the weights equal to the inverse of the uncertainties. Our Eulerien approach relies on the hypothesis that the bloom evolves coherently in the background (resp. over fronts) within
each latitudinal band, which is suggested by high-resolution models of the bloom (e.g. Lévy et al., 2005; Karleskind et al., 2011). It is imperfect as the bloom evolves along Lagrangian trajectories, but provided very consistent results.

3 Results

3.1 Distribution of fronts

Our definition of weak fronts and strong fronts is primarily based on thresholds, derived from the HI distributions (Fig. 3, red dashed line). In each biome, the majority of HI values are below 5, and as mentioned before, we used this threshold to distinguish pixels in the background from those over fronts (when HI > 5). Secondly, the number of points in the HI distribution decreases sharply as the HI value increases above 5, reflecting the fact that fronts with stronger SST gradients are much less frequent than fronts with weaker gradients, as one would expect. We used the HI threshold of 10 to distinguish weak fronts from strong fronts. This separation is imperfect, as seen for instance in Fig. 2 where a few pixels with HI values larger than 10 appear at the acre of an otherwise weak front

230 appear at the core of an otherwise weak front.

However, the choice of these two HI thresholds is also guided by the resulting global spatial climatology of weak and strong fronts (Fig. 4). Weak fronts are abundant, and more or less evenly distributed, over a broad band around and north of the Gulf



Figure 4. Occurrence of (a) weak fronts and (b) strong fronts expressed as the percentage of time steps over the time series (2000–2020) for which a given pixel is occupied by a front.

Stream jet (Fig. 4a). To the south of the Gulf Stream jet, weak fronts are less present, with nevertheless more fronts on the edges of the subtropical gyre (around 28°N) than in its center. This distribution of weak fronts is consistent with the predominance of mesoscale variability observed along the Gulf Stream system from satellite altimetry (Zhai et al., 2008), and the injection of eddy kinetic energy north of the Gulf Stream jet by the Gulf Stream extension. It is thus consistent with the generation process of ephemeral fronts through mesoscale strain.

The climatological distribution of strong fronts shows that they coincide mainly with the two persistent fronts of the Western Boundary Current system, the Gulf Stream front extending northeast from Cape Hatteras, and the more northerly shelf-break front extending eastward to 50°W, following the northeastern U.S. continental shelf break (Fig. 4b). Thus these contrasted coverage of localized strong fronts and more widespread weak fronts are consistent with the hypothesis that weak fronts capture the ephemeral fronts, whereas strong fronts capture the persistent ones.

With the chosen thresholds, the areal proportion of weak fronts tends to increase from South to North, with 7%, 19% and 42% of HI values comprised between 5 and 10 in the permanent subtropical, seasonal subtropical, and subpolar biome, respectively.
Regarding strong fronts, there are only present in the seasonal subtropical and subpolar biome, consistent with the fact that the Gulf Stream and shelf-break fronts are located within these biomes, where HI values above 10 account for 6% and 17%, respectively.

The fraction of the area occupied by fronts also varies with seasons, with generally fewer fronts in summer (Fig. 5d-f). In the permanent subtropical biome, weak fronts cover on average up to 12% in spring and drop to 2% in summer. In the seasonal subtropical biome, the variation is from 27% to 13% for weak fronts, and 8% to 4% for strong fronts. In the subpolar biome, strong fronts cover between 11% in summer and 26% at their peak. This seasonality is consistent with other estimates that submesoscale frontal activity is greater in winter due to greater mixed-layer thickness compared to summer (Callies et al., 2015), and also consistent with the most recent modelling results by Dong et al. (2020) in another Western Boundary system (the Kuroshio Extension) where they show that the strongest submesoscale dynamics occur with a lag of about a month after the mixed layer thickness maximum is reached.

3.2 Local Chl-*a* excess over fronts

The local Chl-*a* excess over fronts is seen for example in Fig. 2, where the highest values of Chl-*a* are found within the HI contour delimiting the front. To quantify this excess over a large number of fronts, we computed for each biome the Chl-*a* distribution sorted by bins of HI (of width 0.1) for the whole time series (Fig. 3). For low values of the heterogeneity index, these distributions are representative of background conditions and reflect the expected differences between biomes: the median Chl-*a* is lowest $(0.05mgm^{-3})$ in the permanent subtropical biome, intermediate in the seasonal subtropical biome $(0.1mgm^{-3})$, and highest in the subpolar biome $(0.2mgm^{-3})$. The Chl-*a* variability along the year and within the biome (i.e. the width of the distribution, highlighted by grey shading in Fig. 3) is larger moving northwards, because of the higher seasonal variability. Importantly, in all 3 biomes, Chl-*a* values increase with HI, and are also more dispersed as HI increases (Fig. 3). Thus the excess in Chl-*a* depends continuously on the value of HI.

In the permanent subtropical biome (Fig. 5a), the seasonal variations of Chl-*a* are very modest with a weak peak in winter. In the seasonal subtropical biome (Fig. 5b), the seasonal variations are well marked, with a minimum in summer, and an increase that starts in fall and peaks in late winter. In the subpolar biome (Fig. 5c), there is a marked bloom in spring with a peak in Chl-*a* in April, followed by an autumn bloom albeit with a smaller magnitude. These different phenologies are

- 270 well documented and largely explained by the differences in the seasonal cycle of the mixed-layer, and the relative depths of the winter mixed-layer and the nutricline (see for instance Lévy et al. (2005) for a description of the drivers of these three production regimes in similar biomes of the Northeast Atlantic). In the permanent subtropical biome, the low productivity is due to the fact that winter mixing is not sufficient to provide a substantial convective supply of nutrients; in contrast, in the seasonal subtropical regime, the increase in production in fall starts when the mixed-layer deepens and reaches the nutricline,
- 275 leading to a fall-winter bloom. In the subpolar biome, this fall bloom is interrupted in winter when the mixed-layer significantly deepens, diluting phytoplankton cells vertically, and a spring bloom is initiated when the mixed-layer stratifies at the end of winter.

In both subtropical biomes, Chl-*a* over fronts is systematically larger than in the background, and this holds throughout the year (Fig. 5a-b). This increase is very modest over the permanent subtropical biome. The local increase over weak fronts also remains modest in the seasonal subtropical gyre, but is much larger over strong fronts. Finally in the subpolar biome (Fig. 5c), Chl-*a* is significantly higher over fronts from the fall until the peak of the bloom but this difference diminishes as the spring bloom decays, and throughout summer. We can also note that the standard deviation range of the yearly median values (Fig. 5a-c) is smaller than the differences between fronts and background, which further confirms that these differences are robust over 20 years of data, for the three biomes and the two types of fronts.

Within each biome, the strength of the local excess in Chl-*a* over fronts varies with latitude. In the permanent subtropical biome (Fig. 6), the excess is only detectable north of 25°N and remains modest (E < 10%); it reaches its maximum of 10% at the beginning and end of the production season. In the seasonal subtropical biome (Fig. 7), the excess is also small in the



Figure 5. Seasonal climatologies in the permanent subtropical biome (1st column), in the seasonal subtropical biome (2nd column) and in the subpolar biome (3rd column). (a-b-c) Chl-*a* median values (top row) over weak fronts (blue), strong fronts (green) and background (red). Gaps between the curves represent the local excess over fronts. (d-e-f) Surface fraction occupied by fronts. (g-h-i) Regional Chl-*a* surplus at the scale of the biome. The surplus accounts for the local excess and for the number of fronts (see method). The plain lines represent the climatological mean, and the envelopes mark the standard deviation over the period 2000–2020. The excess in Chl-*a* is larger over strong fronts than over weak fronts, but weak fronts are more numerous than strong fronts, resulting in a Chl-*a* surplus that can be comparable or even larger for weak fronts.



Figure 6. Permanent subtropical biome: local Chl-*a* excess over fronts by range of latitudes. (a-c-e) Chl-*a* median values over weak fronts (blue) and background (red), (b-d-f) corresponding local excess of Chl-*a* in weak fronts computed as the relative difference of Chl-*a* in fronts and in the background. The plain lines represent the climatological mean, and the envelopes mark the standard deviation over the period 2000–2020. The excess from south to north.

southern part of the biome (south of 35° N) but strongly increases further north (between 35° N and 45° N) and decreases again going even further North (between 40° N and 45° N). Moreover, in this biome, the increase is stronger in summer (July–August) compared to winter (December–January). Finally in the subpolar biome, the magnitude of the local excess diminishes going northward. It tends to be stronger during winter and during the bloom, and weaker in summer. In fact in summer, the differences are hardly discernible when averaged over the entire biome (Fig. 5c), but there is an excess Chl-*a* in the southern part of the biome (<45°N), and a small deficit in the northern part of the biome (>45°N) (Fig. 8). It ensues that the annual mean local excess in Chl-*a* over fronts has a distinct latitudinal pattern (Fig.9a). The strongest local increase due to fronts is located in the

290

295 latitudinal band 35–45°N, where the seasonal subtropical and subpolar biomes meet. The excess is two to three times larger over fronts North of the Gulf Stream (i.e. in the subpolar biome) than South of it (i.e. in the seasonal subtropical biome). Overall, the annual mean local excess over weak fronts varies between 0 and +30%, and between 15% and +60% over strong fronts.



Figure 7. Seasonal subtropical biome: local Chl-*a* excess over fronts by range of latitudes. (a-c-e) Chl-*a* median values over weak fronts (blue), strong fronts (green) and background (red), (b-d-f) corresponding local excess of Chl-*a* in weak and strong fronts computed as the relative difference of Chl-*a* in fronts and in the background. The plain lines represent the climatological mean, and the envelopes mark the standard deviation over the period 2000–2020. The excess is maximum at mid-latitudes.

3.3 Biome-scale Chl-a surplus associated with fronts

- 300 Within each biome, the surplus Chl-*a* associated with the presence of fronts accounts both for the relative Chl-*a* excess over fronts and for the relative area covered by fronts. In the permanent subtropical biome, the surplus associated with weak fronts varies between a maximum of 3%, and a minimum of zero in summer when the coverage of fronts is the lowest (Fig. 5g). The magnitude of the surplus is larger in the seasonal subtropical biome where it reaches 7% in May for weak fronts (Fig. 5h). We can note that the surplus associated with strong fronts is much weaker despite a much greater local impact of strong fronts
- 305 (Fig. 5b) due to the small surface area covered by strong fronts (Fig. 5e). The largest surplus is found in the subpolar biome with maximum values of 12% and due to strong fronts in March. In contrast to the seasonal subtropical biome, in the subpolar biome the surplus associated to strong fronts is larger than the surplus associated to weak fronts. Interestingly our results also reveal a Chl-*a* deficit associated with fronts during the decay phase of the bloom and in summer (negative surplus, Fig. 5i), explained by the negative excess in the northern part of the subpolar biome (Fig. 8b-d).
- 310 Overall, the annual mean Chl-*a* surplus due to fronts is very modest (Fig.9b). All fronts added, the surplus is of the order of 1% for the permanent subtropical biome and 6% for the seasonal subtropical and subpolar biomes, with contributions from



Figure 8. Subpolar biome: local Chl-*a* excess over fronts by range of latitudes in the biome. (a-c-e) Chl-*a* median values over weak fronts (blue), strong fronts (green) and background (red), (b-d-f) corresponding local excess of Chl-*a* in weak and strong fronts computed as the relative difference of Chl-*a* in fronts and in the background. The plain lines represent the climatological mean, and the envelopes mark the standard deviation over the period 2000–2020. The excess decreases from south to north.



Figure 9. (a) Annual mean local Chl-*a* excess over fronts (in %), sorted by latitudinal band (x-axis), by biome (shape of symbol) and by front type (weak fronts in blue, strong fronts in green). (b) Annual mean global surplus of Chl-*a* (in %) for each biome, sorted by front type.

weak and strong fronts which have similar magnitudes. Therefore, there is roughly 5% more Chl-*a* in the Gulf Stream region than there would be in the absence of fronts.

3.4 Bloom timing in the subpolar biome

- Another strong impact of fronts is the earlier onset of the bloom over fronts in the subpolar biome (Fig. 5c). The spring bloom onset propagates from South to North in the biome, starting in early April at 35°N and in late June at 55°N (Fig. 8). In all latitudinal bands, we found that the bloom onset occurs one week earlier over weak fronts than in the background (by -6.7±1.1 days) and two weeks earlier over strong fronts (by -13.5±1.5 days) (Fig. 10). There is a large spread in bloom onset dates in the 20 years of data, due to the very intermittent nature of the bloom onset (Keerthi et al., 2021) that makes it difficult to detect with precision during certain years. Furthermore, in many cases no difference in bloom onset or duration could be detected between the fronts and the background (dots aligned on the diagonal in Fig. 10). Nevertheless, for individual
 - years, delays larger than one month could occur.

4 Discussion

325

Our analyses of surface satellite data over the open ocean in the Gulf Stream extension region, based on the computation of an heterogeneity index HI, allowed us to show a substantial local excess of surface Chl-*a* concentrations over SST fronts compared to background levels, to detect earlier blooms by one to two weeks over fronts, and to quantify that the regional surplus in surface Chl-*a* at the scale of the region associated with fronts was less than 5%. The background levels in this region are very contrasted seasonally and geographically, with a productive and highly seasonal subpolar biome North of the Gulf



Figure 10. Subpolar biome. Comparison of bloom onset dates (in day of year) in the background (x-axis) and over fronts (y-axis), sorted by strength of fronts (shape of symbol), and latitudinal band (color). The line y = x is plotted in black. The distance between the black line and the dotted (respectively dashed) grey line is the measure of the average difference between weak (respectively strong) fronts and background. The bloom onset day propagates from south to north and starts earlier over fronts at all latitudes in the subpolar biome.

330

Stream, a more steady oligotrophic permanent subtropical biome to the South, and an intermediate situation in between the two, where a seasonal subtropical biome prevails. The main results above hold for these three contrasted biomes, although with different intensities, which also depend on the strength of HI.

4.1 Caveats

The level 4 SST product used in this study has the advantage to be readily available on download platforms (here CMEMS) at a reasonably high spatial resolution (4km), avoiding the need to regrid as was done in Liu and Levine (2016) who used 1km resolution L2 MODIS-aqua data. It also has an excellent spatial coverage (e.g. Fig. 1) as it includes merged data from several sensors, but there is a trade-off between coverage and resolution. We have performed initial tests to investigate the differences in using 1km and 4km resolution SST data, which convinced us that the use of 4km resolution data was appropriate (not shown), particularly since the heterogeneity index HI is computed here over boxes of 30km × 30km. Nevertheless a more in-depth study of the sensitivity of our results to the resolution of the satellite products could be carried out in the future.

340 Another caveat of this level 4 product is that the spatial interpolation performed to merge data from several sensors smooths out the finer features, particularly when some of the data are obstructed by clouds. Here we somehow avoided these smoothed areas by using only cloud-free Chl-a pixels. Nevertheless, a bias remains in that there may be a positive correlation between areas with fronts and the presence of clouds. This is the case over the Gulf Stream jet, where dramatic surface temperature gradients are found, and constant clouds are detected over the front. Similar effects can be expected over smaller, short-lived 345 fronts, but probably on a smaller scale.

Our evaluation of the regional surplus in Chl-a associated with fronts assumes that the local Chl-a excess due to fronts is only located over fronts. However, the localization of the highest Chl-a do not perfectly coincide with elevated values of HI: they are places where the HI is large and Chl-a is small, and there are also places with elevated patches of Chl-a outside HI contours (see for example in Fig. 2). The apparent mismatch between the exact location of fronts at a given time and the exact

350 location of the Chl-a response to frontal dynamics at the same time may be due to different factors, such as the very dynamic nature of fronts which can lead to the chaotic advection of phytoplankton (and/or nutrients) outside of fronts (Abraham, 1998), the time-scale needed for phytoplankton to respond to nutrient supplies at fronts, or trophic interactions at the front (as shown, for instance, in the modelling study of Mangolte et al., 2022). This implies that part of the excess Chl-a attributable to frontal activity may actually be located outside of fronts. Our results show that the median distributions of Chl-a significantly differ over fronts and outside of fronts, which suggests that most of the signal due to fronts actually occurs over fronts, nevertheless 355 our estimate of the regional impact of fronts is likely an underestimate due to the signal located outside of fronts.

Moreover, our assessment of the effect of fronts on phytoplankton, based on surface Chl-a, is probably a lower estimate given that the episodic nutrient injections due to submesoscale vertical velocities at fronts can get consumed before reaching the surface (Johnson et al., 2010), leading to phytoplankton enhancements that often does not reach the surface and are more intense at sub-surface (Mouriño et al., 2004; Ruiz et al., 2019). In addition, there may be photo-inhibition which prevents phytoplankton to be near the surface in oligotrophic regions, even though they may be impacted by frontal motions.

Finally, the ratio of Chl-a to total phytoplankton biomass in carbon, Chl-a:C, changes under varying environmental conditions and community changes (Behrenfeld et al., 2015; Halsey and Jones, 2015; Inomura et al., 2022). Diatoms exhibit higher Chl-a: C ratios and are more prevalent in fronts and thus would tend to make our biomass surplus estimation overestimated (Tréguer et al., 2018). This uncertainty could be restricted by taking advantage of recent advances in synoptic estimations of

365

360

the phytoplankton functional types concentrations (El Hourany et al., 2019).

We should also note that our estimates are sensitive to the method used to detect fronts. Here, we used a 30 km wide window to compute HI, while Liu and Levine (2016) used a 10 km wide window. We used a wider window in order to detect the wider western boundary current fronts. But finer ephemeral fronts dominate in oligotrophic gyres. Thus the phytoplankton signal at

370 fronts may be underestimated with the use of a wider window there. This difference may explain why our assessment of the excess Chl-a in the North Atlantic subtropical gyre is three times lower than the assessment of Liu and Levine (2016) in the North Pacific subtropical gyre. Dedicated sensitivity experiments would be needed to assess this with more certainty.

4.2 Local Chl-a excess over fronts

With these caveats in mind, we find that the degree of local Chl-a excess over fronts varied seasonally, but mostly varied from

one biome to another, with an intensity which was weaker in the more oligotrophic region, stronger between 35–45°N, and intermediate in the subpolar biome (Fig. 9a). Moreover, the local excess of Chl-*a* was always significantly larger over strong fronts than over weak fronts. The increase in Chl-*a* with increasing front strength (i.e. with increasing HI, Fig. 3) is consistent with the hypothesis that phytoplankton production is amplified at fronts by an enhancement of the flux of nutrients, and that this flux is stronger the stronger the front is. Strong fronts in this study coincide with the persistent fronts of the Western
boundary current system, which are both associated with large and deep reaching vertical velocities (Liao et al., 2022) and with strong lateral transport of nutrients (Pelegrí et al., 1996). On the other hand, the larger dispersion in the Chl-*a* distribution with increasing HI reflects the fact that not all fronts are equally efficient.

Co-occurrence between frontal vertical velocities (or divergence) and enhanced Chl-*a* has been observed over specific fronts in the North Atlantic (Mouriño et al., 2004; Allen et al., 2005; Lehahn et al., 2007). The only study that has statistically
connected enhanced Chl-*a* with the presence of temperature front was conducted in the North Pacific subtropical gyre (Liu and Levine, 2016), which shares characteristics with the permanent subtropical biome examined here. Our results thus extend those of Liu and Levine (2016) to a region with stronger biological contrasts and phenologies, and with more complex dynamics.

One of the factors determining the magnitude of the local Chl-*a* excess over fronts is the magnitude of the vertical nutrient flux, which itself depends on the magnitude of the vertical velocities, of their depth penetration, and of the depth of the nutricline. The nutricline depth shows a sharp latitudinal gradient within this region, from 150m depth at 25°N to 50 m at 50°N (Romera-Castillo et al., 2016). This can explain the maximum magnitude of the Chl-*a* response at the northern edge of the subtropical gyre, where the lack of nutrients is more severely controlling phytoplankton abundance than further north, and where the nutricline is closer to the surface than further south. Another factor than can explain this signal is the nutrient stream, that feeds this intermediate regions with nutrients.

- 395 Moreover, vertical velocities associated with ephemeral fronts, often confined to the mixed layer, are likely to be a less efficient nutrient flux pathway to the euphotic zone from the interior than deep, dynamic, persistent fronts extending well below the mixed layer (Lévy et al., 2018). The contrasting impacts of deep and shallow fronts are striking in models (Lévy et al., 2012), but are difficult to quantify from a small number of in situ observations. Here we observed that the magnitude of the Chl-*a* response over fronts increased with the strength of the heterogeneity index HI (Fig. 3). In other words, strong fronts,
- 400 characterized by high values of HI (HI > 10), led to a stronger increase in Chl-*a* values than weak fronts, characterized by intermediate values of HI (5 < HI < 10).

Finally, an important outcome of this study is that the surplus in phytoplankton biomass associated with fronts can be stronger in blooming biomes than in oligotrophic ones. In fact in regions where nutrients are not limiting productivity, fronts have been shown to subduct excess nutrients (Oschlies, 2002; Gruber et al., 2011) and excess biomass (Lathuilière et al., 2010), rather than to lead to an increase in biomass as in oligotrophic regions. This effect of a decreased biomass, suggestive of subduction, is observed here in the northern parts of the subpolar biome between May and September (Fig. 8b-d). But we also find that in

405

the subpolar biome, the Chl-*a* excess over fronts can reach 150% during the bloom and 50% during summer (Fig. 8), while in the permanent subtropical biome it never exceeds 10% (Fig. 6). The enhancement of the spring bloom by submesoscale frontal processes observed here was recently also put forward in a modelling study by Simoes-Sousa et al. (2022).

410 4.3 Persistent and ephemeral fronts

The above results are suggestive that the heterogeneity index could be used as a way to discriminate between persistent (and deep) fronts, and ephemeral (and shallower) fronts. The localization and frequency of strong and weak fronts is consistent with this hypothesis. Weak fronts are much more frequent than strong fronts, as we expect from ephemeral fronts compared with persistent ones (Fig. 4). In addition, the localization of strong fronts coincide with the position of the Gulf Stream and shelf-break front. Another element that supports this hypothesis is the scale over which the HI is computed (30km) which gives a strong weight to SST heterogeneities associated with large contrasts which is the case across the Gulf Stream and shelf-break. Of course, more direct evidence linking the penetration of fronts with the intensity of the heterogeneity index would be needed

to confirm the association.

415

4.4 Biome-scale Chl-a amplification associated with fronts

- 420 The categorization of fronts based on HI has allowed us to quantify the respective contribution of two types of fronts on the regional Chl-*a* amplification (Fig. 9b). Weak fronts are associated with a local Chl-*a* excess which is weaker than strong front, in general, but because they are also more frequent than strong fronts, depending on the biome and seasons, they contribute equally to the regional Chl-*a* surplus as strong fronts. There is also some degree of seasonality in this small surplus of Chl-*a* attributed to fronts, which heavily depends on the region of interest (Fig. 5). As predicted by theory and noted by previous
- 425 studies, sub-mesoscale fronts which are confined to the mixed-layer are less abundant in summer when mixed-layers are shallower. In the south zone, this leads to an overall weaker effect of fronts in summer (near 0%) relative to the rest of the year (less than 3% average), but in the jet area, it is compensated by a larger intensity of the increase in Chl-*a* in summer leading to a Chl-*a* surplus in summer (7% for weak fronts) which is much larger than in winter (1%). In the north, the situation is quite different with an impact of fronts close to zero during the spring bloom, negative in summer as vertical velocities at fronts are also capable of sinking the surface bloom (Lévy et al., 2018), and maximal in autumn and winter.
 - Besides these small spatial and temporal variations in amplitude, a key result of this study is that despite strong local impact of fronts, their overall contribution at large-scale remains small, a few percent at most, and of the order of 5% for the entire region. Nevertheless, this result should be considered as a lower bound, first because increases in Chl-*a* at fronts are often stronger at subsurface than at the surface, and second because in a region characterized by strong gradients like this one,
- 435 additional nutrient fluxes due to frontal activity might not necessarily lead to local anomalies in Chl-*a*, but could also be hidden by the large-scale gradient. Finally, 5% amplification of surface Chl-*a* might lead to greater amplification at higher trophic level (Stock et al., 2014; Lotze et al., 2019), with ecological implications that remain to be evaluated.

4.5 Earlier blooms over fronts

Another key result of this study is the detection of earlier blooms over fronts than over background conditions in the north

- 440 of the Gulf Stream jet. Several field and modeling studies have shown that frontal dynamics, by tilting existing horizontal density gradients, increase the vertical stratification of surface mixed layers (Taylor and Ferrari, 2011), which can lead to the stratification of the mixed layer prior to seasonal stratification. Given that the surface spring bloom is triggered by increased stratification, this effect can cause earlier local phytoplankton blooms over fronts compared to surrounding areas.
- However, while the increased stratification over fronts can be directly observed in situ (Karleskind et al., 2011; Mahadevan et al., 2012), how it affects the timing of the bloom has so far been quantified with numerical models only, due to the difficulty in tracking the bloom evolution over fronts which themselves evolve over time (Lévy et al., 2000; Karleskind et al., 2011; Mahadevan et al., 2012). We provide here the first observational evidence of the early onset of blooms over fronts. Moreover, our estimate leads to smaller values (earlier blooms by one to two weeks) than previously estimated from models (20–30 days by Mahadevan et al. (2012)).
- The method that we used to quantify differences in bloom timing over fronts and background is based on the time evolution of an eulerian quantity, the Chl-*a* median over latitudinal bands, whereas the bloom evolves along lagrangian trajectories. Considering a rather small area, as we have done here, is a way of overcoming the difficulty of following the temporal evolution on fronts whose life history is too complex to be captured and shorter than the bloom itself. It also limits the impact that the northward propagation of the bloom could have on the temporal assessment. It should also be noted that it is inherently
- difficult to pinpoint the precise onset and end days of a bloom, as the spring bloom shows large intraseasonal variability in its characteristics; its beginning can be more or less sudden, and is often made of multiple peaks (Keerthi et al., 2020).

5 Conclusions

The open ocean Gulf Stream extension region is a region of strong biological contrasts and particularly strong frontal activity of the world's ocean, undergoing rapid warming which strongly affect fisheries (Pershing et al., 2015; Neto et al., 2021).
Quantifying the impact of fronts on phytoplankton there is thus particularly relevant, and we expected to detect a large impact. The use of 20 years of satellite data of SST to detect fronts and of surface Chl-*a* to compute anomalies over the front allowed us to provide a robust assessment of this impact. We found three main results. First, that the regional increase in surface phytoplankton associated with fronts is rather modest, 5% at most; second that nutrient supplies at fronts enhanced the spring bloom two to three three times more than they enhanced oligotrophic regions; and third, that the spring bloom onset was earlier

465 over fronts by one to two weeks, which we already knew from models (Karleskind et al., 2011; Mahadevan et al., 2012) but for which we had no direct evidence nor sound quantification. We also showed a reduction of phytoplankton over fronts at the end of the bloom, that we attributed to subduction.

Although limited to the Gulf Stream region, this study provides a well-tested methodology that could enable the study of the links between small-scale ocean physics and phytoplankton response in other regions of the global ocean. In addition, these results on the importance of fronts for phytoplankton biomass and phenology could also be used to evaluate models



Figure A1. Climatological mean of Chl-*a* median values (top row) over weak fronts (blue), strong fronts (green) and background (red), surface fraction occupied by weak fronts and strong fronts (middle row), and global Chl-*a* excess due to weak and strong fronts (bottom row). Each line represent a set of parameter with the bolder line indicating the retained set of parameters. The tested rolling window sizes are 20km, 30km and 40km. Different normalization coefficients are tested for a 30km window size: double the variance, double the bimodality, and double the skewness.

coupling ocean physics and phytoplankton at high spatial resolution, or to test parameterizations representing the effect of small scales on phytoplankton production in coarser resolution models. Finally, the combination of these observation-based results with theoretical arguments and well-assessed models should also allow us to better constrain the response of phytoplankton production to climate change (Couespel et al., 2021), which still has very large uncertainties as shown by the latest set of Earth system models (Kwiatkowski et al., 2020).

Code availability. All the scripts needed to reproduce our results, as well as the data necessary to generate the figures in this manuscript are available on a Zenodo repository (DOI: 10.5281/zenodo.7470199, Haëck et al. (2022)).



Figure A2. Distribution of Chl-*a* of the year 2007 by seasons (rows), for the three biomes (columns), and for the background (red), weak fronts (blue) and strong fronts (green). The median value of each distribution is indicated by a vertical line.

Author contributions. ML, LB and CH conceived the study. CH conceived the methodology and performed the analysis. ML and CH wrote the paper. All authors contributed to the analysis and discussion of the results.

480 *Competing interests.* The authors declare that they have no conflict of interest.

Acknowledgements. CH benefited from a PhD scholarship by ENS. The project was supported by TOSCA CNES and by the ENS CHANEL chair. We thank Daniele Iudicone, Francesco d'Ovidio, Sakina-Dorothée Ayata, and Amala Mahadevan for the useful discussions which helped to refine our methodology. We thank Xioa Liu for helping us reproduce their work.

This study has been conducted using E.U. Copernicus Marine Service Information (datasets used: Garnesson et al., 2021; Good et al., 2020b).

GlobColour data (https://globcolour.info) used in this study has been developed, validated, and distributed by ACRI-ST, France.

References

Abraham, E. R.: The Generation of Plankton Patchiness by Turbulent Stirring, Nature, 391, 577–580, https://doi.org/10.1038/35361, 1998.

- Allen, J. T., Brown, L., Sanders, R., Mark Moore, C., Mustard, A., Fielding, S., Lucas, M., Rixen, M., Savidge, G., Henson,
 S., and Mayor, D.: Diatom Carbon Export Enhanced by Silicate Upwelling in the Northeast Atlantic, Nature, 437, 728–732, https://doi.org/10.1038/nature03948, 2005.
 - Behrenfeld, M. J., O'Malley, R. T., Boss, E. S., Westberry, T. K., Graff, J. R., Halsey, K. H., Milligan, A. J., Siegel, D. A., and Brown, M. B.: Revaluating Ocean Warming Impacts on Global Phytoplankton, Nat. Clim. Change, 6, 323, https://doi.org/10.1038/nclimate2838, 2015.

Belkin, I. M. and O'Reilly, J. E.: An Algorithm for Oceanic Front Detection in Chlorophyll and SST Satellite Imagery, J. Marine Syst., 78,

- 495 319–326, https://doi.org/10.1016/j.jmarsys.2008.11.018, 2009.
 - Belkin, I. M., Cornillon, P. C., and Sherman, K.: Fronts in Large Marine Ecosystems, Prog. Oceanogr., 81, 223–236, https://doi.org/10.1016/j.pocean.2009.04.015, 2009.
 - Bock, N., Cornec, M., Claustre, H., and Duhamel, S.: Biogeographical Classification of the Global Ocean from BGC-Argo Floats, Global Biogeochem. Cy., 36, e2021GB007 233, https://doi.org/10.1029/2021GB007233, 2022.
- 500 Calil, P. H. R., Doney, S. C., Yumimoto, K., Eguchi, K., and Takemura, T.: Episodic Upwelling and Dust Deposition as Bloom Triggers in Low-Nutrient, Low-Chlorophyll Regions, J. Geophys. Res., 116, C06 030, https://doi.org/10.1029/2010JC006704, 2011.
 - Callies, J., Ferrari, R., Klymak, J. M., and Gula, J.: Seasonality in Submesoscale Turbulence, Nat. Commun., 6, 6862, https://doi.org/10.1038/ncomms7862, 2015.
 - Cayula, J.-F. and Cornillon, P.: Edge Detection Algorithm for SST Images, J. Atmos. Oceanic Tech., 9, 67–80, https://doi.org/10.1175/1520-
- 505 0426(1992)009<0067:edafsi>2.0.co;2, 1992.
- Contreras, M., Renault, L., and Marchesiello, P.: Understanding Energy Pathways in the Gulf Stream, J. Phys. Oceanogr., 53, 719–736, https://doi.org/10.1175/JPO-D-22-0146.1, 2023.
 - Couespel, D., Lévy, M., and Bopp, L.: Oceanic Primary Production Decline Halved in Eddy-Resolving Simulations of Global Warming, Biogeosciences, 18, 4321–4349, https://doi.org/10.5194/bg-18-4321-2021, 2021.
- 510 Doney, S. C., Glover, D. M., McCue, S. J., and Fuentes, M.: Mesoscale Variability of Sea-viewing Wide Field-of-view Sensor (SeaWiFS) Satellite Ocean Color: Global Patterns and Spatial Scales, J. Geophys. Res. Oceans, 108, https://doi.org/10.1029/2001jc000843, 2003.
 - Dong, J., Fox-Kemper, B., Zhang, H., and Dong, C.: The Seasonality of Submesoscale Energy Production, Content, and Cascade, Geophys. Res. Lett., 47, https://doi.org/10.1029/2020GL087388, 2020.
 - Drushka, K., Asher, W. E., Sprintall, J., Gille, S. T., and Hoang, C.: Global Patterns of Submesoscale Surface Salinity Variability, J. Phys.
- 515 Oceanogr., 49, 1669–1685, https://doi.org/10.1175/jpo-d-19-0018.1, 2019.
 - El Hourany, R., Abboud-Abi Saab, M., Faour, G., Aumont, O., Crépon, M., and Thiria, S.: Estimation of Secondary Phytoplankton Pigments from Satellite Observations Using Self-Organizing Maps (SOMs), J. Geophys. Res. Oceans, 124, 1357–1378, https://doi.org/10.1029/2018jc014450, 2019.
 - Flagg, C. N., Dunn, M., Wang, D.-P., Rossby, H. T., and Benway, R. L.: A Study of the Currents of the Outer Shelf and
- 520 Upper Slope from a Decade of Shipboard ADCP Observations in the Middle Atlantic Bight, J. Geophys. Res. Oceans, 111, https://doi.org/10.1029/2005JC003116, 2006.

- Garnesson, P., Mangin, A., Fanton d'Andon, O., Demaria, J., and Bretagnon, M.: The CMEMS GlobColour Chlorophyll *a* Product Based on Satellite Observation: Multi-Sensor Merging and Flagging Strategies, Ocean Sci., 15, 819–830, https://doi.org/10.5194/os-15-819-2019, 2019.
- 525 Garnesson, P., Mangin, A., Fanton d'Andon, O., Demaria, J., and Bretagnon, M.: Global Ocean Colour (Copernicus-GlobColour), Bio-Geo-Chemical, L3 (daily) from Satellite Observations (1997-ongoing), E.U. Copernicus Marine Service Information [dataset], https://doi.org/10.48670/moi-00280, 2021.
 - Glover, D. M., Doney, S. C., Oestreich, W. K., and Tullo, A. W.: Geostatistical Analysis of Mesoscale Spatial Variability and Error in SeaWiFS and MODIS/Aqua Global Ocean Color Data, J. Geophys. Res. Oceans, 123, 22–39, https://doi.org/10.1002/2017JC013023, 2018.

530 20

- Good, S., Fiedler, E., Mao, C., Martin, M. J., Maycock, A., Reid, R., Roberts-Jones, J., Searle, T., Waters, J., While, J., and Worsfold, M.: The Current Configuration of the OSTIA System for Operational Production of Foundation Sea Surface Temperature and Ice Concentration Analyses, Remote Sensing, 12, 720, https://doi.org/10.3390/rs12040720, 2020a.
 - Good, S., Fiedler, E., Mao, C., Martin, M. J., Maycock, A., Reid, R., Roberts-Jones, J., Searle, T., Waters, J., While, J., and Worsfold,
- 535 M.: ESA SST CCI and C3S reprocessed sea surface temperature analyses, E.U. Copernicus Marine Service Information [dataset], https://doi.org/10.48670/moi-00169, 2020b.
 - Gruber, N., Lachkar, Z., Frenzel, H., Marchesiello, P., Münnich, M., McWilliams, J. C., Nagai, T., and Plattner, G.-K.: Eddy-Induced Reduction of Biological Production in Eastern Boundary Upwelling Systems, Nat. Geosci., 4, 787–792, https://doi.org/10.1038/ngeo1273, 2011.
- 540 Guo, M., Xiu, P., Chai, F., and Xue, H.: Mesoscale and Submesoscale Contributions to High Sea Surface Chlorophyll in Subtropical Gyres, Geophys. Res. Lett., 46, 13 217–13 226, https://doi.org/10.1029/2019gl085278, 2019.
 - Gupta, M., Williams, R. G., Lauderdale, J. M., Jahn, O., Hill, C., Dutkiewicz, S., and Follows, M. J.: A Nutrient Relay Sustains Subtropical Ocean Productivity, P. Natl. Acad. Sci. USA, 119, e2206504 119, https://doi.org/10.1073/pnas.2206504119, 2022.

Halsey, K. H. and Jones, B. M.: Phytoplankton Strategies for Photosynthetic Energy Allocation, Annu. Rev. Mar. Sci., 7, 265-297,

545 https://doi.org/10.1146/annurev-marine-010814-015813, 2015.

- Hauschildt, J., Thomsen, S., Echevin, V., Oschlies, A., José, Y. S., Krahmann, G., Bristow, L. A., and Lavik, G.: The Fate of Upwelled Nitrate off Peru Shaped by Submesoscale Filaments and Fronts, Biogeosciences, 18, 3605–3629, https://doi.org/10.5194/bg-18-3605-2021, 2021.
- Haëck, C., Lévy, M., Mangolte, I., and Bopp, L.: Satellite Data Reveal Earlier and Stronger Phytoplankton Blooms over Fronts in the Gulf Stream Region: Code and Data, https://doi.org/10.5281/ZENODO.7470199, 2022.
- 550 Inomura, K., Deutsch, C., Jahn, O., Dutkiewicz, S., and Follows, M. J.: Global Patterns in Marine Organic Matter Stoichiometry Driven by Phytoplankton Ecophysiology, Nat. Geosci., 15, 1034–1040, https://doi.org/10.1038/s41561-022-01066-2, 2022.
 - Johnson, K. S., Riser, S. C., and Karl, D. M.: Nitrate Supply from Deep to Near-Surface Waters of the North Pacific Subtropical Gyre, Nature, 465, 1062–1065, https://doi.org/10.1038/nature09170, 2010.

Karleskind, P., Lévy, M., and Mémery, L.: Modifications of Mode Water Properties by Sub-Mesoscales in a Bio-Physical Model of the
 Northeast Atlantic, Ocean Model., 39, 47–60, https://doi.org/10.1016/j.ocemod.2010.12.003, 2011.

Keerthi, M. G., Lévy, M., Aumont, O., Lengaigne, M., and Antoine, D.: Contrasted Contribution of Intraseasonal Time Scales to Surface Chlorophyll Variations in a Bloom and an Oligotrophic Regime, J. Geophys. Res. Oceans, 125, e2019JC015701, https://doi.org/10.1029/2019jc015701, 2020. Keerthi, M. G., Lévy, M., and Aumont, O.: Intermittency in Phytoplankton Bloom Triggered by Modulations in Vertical Stability, Sci. Rep.,

- 560 11, 1285, https://doi.org/10.1038/s41598-020-80331-z, 2021.
 - Keerthi, M. G., Prend, C. J., Aumont, O., and Lévy, M.: Annual Variations in Phytoplankton Biomass Driven by Small-Scale Physical Processes, Nat. Geosci., https://doi.org/10.1038/s41561-022-01057-3, 2022.
- Kessouri, F., Bianchi, D., Renault, L., McWilliams, J. C., Frenzel, H., and Deutsch, C. A.: Submesoscale Currents Modulate the Seasonal Cycle of Nutrients and Productivity in the California Current System, Global Biogeochem. Cy., 34, e2020GB006578, https://doi.org/10.1029/2020GB006578, 2020.
 - Kwiatkowski, L., Torres, O., Bopp, L., Aumont, O., Chamberlain, M., Christian, J. R., Dunne, J. P., Gehlen, M., Ilyina, T., John, J. G., Lenton, A., Li, H., Lovenduski, N. S., Orr, J. C., Palmieri, J., Santana-Falcón, Y., Schwinger, J., Séférian, R., Stock, C. A., Tagliabue, A., Takano, Y., Tjiputra, J., Toyama, K., Tsujino, H., Watanabe, M., Yamamoto, A., Yool, A., and Ziehn, T.: Twenty-First Century Ocean Warming, Acidification, Deoxygenation, and Upper-Ocean Nutrient and Primary Production Decline from CMIP6 Model Projections,
- Biogeosciences, 17, 3439–3470, https://doi.org/10.5194/bg-17-3439-2020, 2020.
 Lathuilière, C., Echevin, V., Lévy, M., and Madec, G.: On the Role of the Mesoscale Circulation on an Idealized Coastal Upwelling Ecosystem, J. Geophys. Res. Oceans, 115, https://doi.org/10.1029/2009JC005827, 2010.
 - Lehahn, Y., d' Ovidio, F., Lévy, M., and Heifetz, E.: Stirring of the Northeast Atlantic Spring Bloom: A Lagrangian Analysis Based on Multisatellite Data, J. Geophys. Res. Oceans, 112, https://doi.org/10.1029/2006jc003927, 2007.
- 575 Letscher, R. T., Primeau, F., and Moore, J. K.: Nutrient Budgets in the Subtropical Ocean Gyres Dominated by Lateral Transport, Nat. Geosci., 9, 815–819, https://doi.org/10.1038/ngeo2812, 2016.
 - Liao, F., Liang, X., Li, Y., and Spall, M.: Hidden Upwelling Systems Associated with Major Western Boundary Currents, J. Geophys. Res. Oceans, 127, https://doi.org/10.1029/2021JC017649, 2022.
 - Linder, C. A. and Gawarkiewicz, G.: A Climatology of the Shelfbreak Front in the Middle Atlantic Bight, J. Geophys. Res. Oceans, 103,

580 18 405–18 423, https://doi.org/10.1029/98JC01438, 1998.

Little, H., Vichi, M., Thomalla, S., and Swart, S.: Spatial and Temporal Scales of Chlorophyll Variability Using High-Resolution Glider Data, J. Marine Syst., 187, 1–12, https://doi.org/10.1016/j.jmarsys.2018.06.011, 2018.

Liu, X. and Levine, N. M.: Enhancement of Phytoplankton Chlorophyll by Submesoscale Frontal Dynamics in the North Pacific Subtropical Gyre, Geophys. Res. Lett., 43, 1651–1659, https://doi.org/10.1002/2015gl066996, 2016.

585 Long, Y., Guo, X., Zhu, X.-H., and Li, Z.: Nutrient Streams in the North Pacific, Prog. Oceanogr., 202, 102756, https://doi.org/10.1016/j.pocean.2022.102756, 2022.

Longhurst, A. R.: Ecological Geography of the Sea, Academic Press, 2nd ed edn., 2007.

Lotze, H. K., Tittensor, D. P., Bryndum-Buchholz, A., Eddy, T. D., Cheung, W. W. L., Galbraith, E. D., Barange, M., Barrier, N., Bianchi, D., Blanchard, J. L., Bopp, L., Büchner, M., Bulman, C. M., Carozza, D. A., Christensen, V., Coll, M., Dunne, J. P., Fulton, E. A.,

- 590 Jennings, S., Jones, M. C., Mackinson, S., Maury, O., Niiranen, S., Oliveros-Ramos, R., Roy, T., Fernandes, J. A., Schewe, J., Shin, Y.-J., Silva, T. A. M., Steenbeek, J., Stock, C. A., Verley, P., Volkholz, J., Walker, N. D., and Worm, B.: Global Ensemble Projections Reveal Trophic Amplification of Ocean Biomass Declines with Climate Change, P. Natl. Acad. Sci. USA, 116, 12907–12912, https://doi.org/10.1073/pnas.1900194116, 2019.
- Lévy, M., Mémery, L., and Madec, G.: Combined Effects of Mesoscale Processes and Atmospheric High-Frequency Variability on the Spring
 Bloom in the MEDOC Area, Deep Sea Research Part I: Oceanogr. Res. Papers, 47, 27–53, https://doi.org/10.1016/s0967-0637(99)00051-
 - Bloom in the MEDOC Area, Deep Sea Research Part I: Oceanogr. Res. Papers, 47, 27–55, https://doi.org/10.1010/s0967-0657(99)00051 5, 2000.

- Lévy, M., Lehahn, Y., André, J.-M., Mémery, L., Loisel, H., and Heifetz, E.: Production Regimes in the Northeast Atlantic: A Study Based on Sea-viewing Wide Field-of-view Sensor (SeaWiFS) Chlorophyll and Ocean General Circulation Model Mixed Layer Depth, J. Geophys. Res. Oceans, 110, https://doi.org/10.1029/2004JC002771, 2005.
- 600 Lévy, M., Ferrari, R., Franks, P. J. S., Martin, A. P., and Rivière, P.: Bringing Physics to Life at the Submesoscale, Geophys. Res. Lett., 39, https://doi.org/10.1029/2012gl052756, 2012.
 - Lévy, M., Franks, P. J. S., and Smith, K. S.: The Role of Submesoscale Currents in Structuring Marine Ecosystems, Nat. Commun., 9, https://doi.org/10.1038/s41467-018-07059-3, 2018.
 - Mahadevan, A.: The Impact of Submesoscale Physics on Primary Productivity of Plankton, Annu. Rev. Mar. Sci., 8, 161-184,
- https://doi.org/10.1146/annurev-marine-010814-015912, 2016.
 Mahadevan, A., D'Asaro, E., Lee, C., and Perry, M. J.: Eddy-Driven Stratification Initiates North Atlantic Spring Phytoplankton Blooms, Science, 337, 54–58, https://doi.org/10.1126/science.1218740, 2012.
 - Mahadevan, A., Pascual, A., Rudnick, D. L., Ruiz, S., Tintoré, J., and D'Asaro, E.: Coherent Pathways for Vertical Transport from the Surface Ocean to Interior, B. Am. Meteorol. Soc., 101, E1996–E2004, https://doi.org/10.1175/BAMS-D-19-0305.1, 2020.
- 610 Mangolte, I., Lévy, M., Dutkiewicz, S., Clayton, S., and Jahn, O.: Plankton Community Response to Fronts: Winners and Losers, J. Plankton Res., 44, 241–258, https://doi.org/10.1093/plankt/fbac010, 2022.
 - Mangolte, I., Lévy, M., Haëck, C., and Ohman, M. D.: Sub-Frontal Niches of Plankton Communities Driven by Transport and Trophic Interactions at Ocean Fronts [Preprint], EGUsphere, https://doi.org/10.5194/egusphere-2023-471, 2023.
 - Marrec, P., Grégori, G., Doglioli, A. M., Dugenne, M., Della Penna, A., Bhairy, N., Cariou, T., Hélias Nunige, S., Lahbib, S., Rougier,
- 615 G., Wagener, T., and Thyssen, M.: Coupling physics and biogeochemistry thanks to high-resolution observations of the phytoplankton community structure in the northwestern Mediterranean Sea, Biogeosciences, 15, 1579–1606, https://doi.org/10.5194/bg-15-1579-2018, 2018.
 - Mauzole, Y.: Objective Delineation of Persistent SST Fronts Based on Global Satellite Observations, Remote Sens. Environ., 269, 112798, https://doi.org/10.1016/j.rse.2021.112798, 2022.
- 620 McWilliams, J. C.: Submesoscale Currents in the Ocean, Proc. R. Soc. A.-Math. Phy., 472, 20160117, https://doi.org/10.1098/rspa.2016.0117, 2016.
 - Merchant, C. J., Embury, O., Bulgin, C. E., Block, T., Corlett, G. K., Fiedler, E., Good, S. A., Mittaz, J., Rayner, N. A., Berry, D., Eastwood, S., Taylor, M., Tsushima, Y., Waterfall, A., Wilson, R., and Donlon, C.: Satellite-Based Time-Series of Sea-Surface Temperature since 1981 for Climate Applications, Scientific Data, 6, 223, https://doi.org/10.1038/s41597-019-0236-x, 2019.
- 625 Mouriño, B., Fernández, E., and Alves, M.: Thermohaline Structure, Ageostrophic Vertical Velocity Fields and Phytoplankton Distribution and Production in the Northeast Atlantic Subtropical Front, J. Geophys. Res. Oceans, 109, https://doi.org/10.1029/2003jc001990, 2004.
 - Neto, A. G., Langan, J. A., and Palter, J. B.: Changes in the Gulf Stream Preceded Rapid Warming of the Northwest Atlantic Shelf, Communications Earth & Environment, 2, 1–10, https://doi.org/10.1038/s43247-021-00143-5, 2021.
- Omand, M. M., D'Asaro, E. A., Lee, C. M., Perry, M. J., Briggs, N., Cetinić, I., and Mahadevan, A.: Eddy-Driven Subduction Exports
 Particulate Organic Carbon from the Spring Bloom, Science, 348, 222–225, https://doi.org/10.1126/science.1260062, 2015.
 - Oschlies, A.: Nutrient Supply to the Surface Waters of the North Atlantic: A Model Study, J. Geophys. Res. Oceans, 107, 14–1–14–13, https://doi.org/10.1029/2000JC000275, 2002.
 - Pelegrí, J. L., Csanady, G. T., and Martins, A.: The North Atlantic Nutrient Stream, J. Oceanogr., 52, 275–299, https://doi.org/10.1007/BF02235924, 1996.

- Pershing, A. J., Alexander, M. A., Hernandez, C. M., Kerr, L. A., Le Bris, A., Mills, K. E., Nye, J. A., Record, N. R., Scannell, H. A., Scott, 635 J. D., Sherwood, G. D., and Thomas, A. C.: Slow Adaptation in the Face of Rapid Warming Leads to Collapse of the Gulf of Maine Cod Fishery, Science, 350, 809-812, https://doi.org/10.1126/science.aac9819, 2015.
 - Romera-Castillo, C., Letscher, R. T., and Hansell, D. A.: New Nutrients Exert Fundamental Control on Dissolved Organic Carbon Accumulation in the Surface Atlantic Ocean, P. Natl. Acad. Sci. USA, 113, 10497-10502, https://doi.org/10.1073/pnas.1605344113, 2016.
- Ruiz, S., Claret, M., Pascual, A., Olita, A., Troupin, C., Capet, A., Tovar-Sánchez, A., Allen, J., Poulain, P.-M., Tintoré, J., and Mahadevan, 640 A.: Effects of Oceanic Mesoscale and Submesoscale Frontal Processes on the Vertical Transport of Phytoplankton, J. Geophys. Res. Oceans, 124, 5999-6014, https://doi.org/10.1029/2019ic015034, 2019.
 - Sanchez-Rios, A., Shearman, R. K., Klymak, J., D'Asaro, E., and Lee, C.: Observations of Cross-Frontal Exchange Associated with Submesoscale Features along the North Wall of the Gulf Stream, Deep Sea Research Part I: Oceanogr. Res. Papers, 163, 103 342,
- 645 https://doi.org/10.1016/j.dsr.2020.103342, 2020.

650

665

- Sarmiento, J. L., Slater, R., Barber, R., Bopp, L., Doney, S. C., Hirst, A. C., Kleypas, J., Matear, R., Mikolajewicz, U., Monfray, P., Soldatov, V., Spall, S. A., and Stouffer, R.: Response of Ocean Ecosystems to Climate Warming, Global Biogeochem. Cy., 18, https://doi.org/10.1029/2003GB002134, 2004.
- Simoes-Sousa, I. T., Tandon, A., Pereira, F., Lazaneo, C. Z., and Mahadevan, A.: Mixed Layer Eddies Supply Nutrients to Enhance the
- Spring Phytoplankton Bloom, Front. Mar. Sci., 9, 825 027, https://doi.org/10.3389/fmars.2022.825027, 2022. Stock, C. A., Dunne, J. P., and John, J. G.: Drivers of Trophic Amplification of Ocean Productivity Trends in a Changing Climate, Biogeosciences, 11, 7125-7135, https://doi.org/10.5194/bg-11-7125-2014, 2014.
 - Taylor, J. R. and Ferrari, R.: Ocean Fronts Trigger High Latitude Phytoplankton Blooms, Geophys. Res. Lett., 38, https://doi.org/10.1029/2011gl049312, 2011.
- 655 Thomas, L. and Ferrari, R.: Friction, Frontogenesis, and the Stratification of the Surface Mixed Layer, J. Phys. Oceanogr., 38, 2501–2518, https://doi.org/10.1175/2008JPO3797.1, 2008.
 - Thomas, L. N., Tandon, A., and Mahadevan, A.: Submesoscale Processes and Dynamics, in: Geophysical Monograph Series, edited by Hecht, M. W. and Hasumi, H., vol. 177, pp. 17–38, American Geophysical Union, https://doi.org/10.1029/177GM04, 2008.
 - Tréguer, P., Bowler, C., Moriceau, B., Dutkiewicz, S., Gehlen, M., Aumont, O., Bittner, L., Dugdale, R., Finkel, Z., Iudicone, D., Jahn, O.,
- 660 Guidi, L., Lasbleiz, M., Leblanc, K., Levy, M., and Pondaven, P.: Influence of Diatom Diversity on the Ocean Biological Carbon Pump, Nat. Geosci., 11, 27-37, https://doi.org/10.1038/s41561-017-0028-x, 2018.
 - Tzortzis, R., Doglioli, A. M., Barrillon, S., Petrenko, A. A., d' Ovidio, F., Izard, L., Thyssen, M., Pascual, A., Barceló-Llull, B., Cyr, F., Tedetti, M., Bhairy, N., Garreau, P., Dumas, F., and Gregori, G.: Impact of Moderately Energetic Fine-Scale Dynamics on the Phytoplankton Community Structure in the Western Mediterranean Sea, Biogeosciences, 18, 6455–6477, https://doi.org/10.5194/bg-18-6455-2021, 2021.
 - Uchida, T., Balwada, D., P. Abernathey, R., A. McKinley, G., K. Smith, S., and Lévy, M.: Vertical Eddy Iron Fluxes Support Primary Production in the Open Southern Ocean, Nat. Commun., 11, 1125, https://doi.org/10.1038/s41467-020-14955-0, 2020.
 - Verneil, A., Franks, P. J. S., and Ohman, M. D.: Frontogenesis and the Creation of Fine-scale Vertical Phytoplankton Structure, J. Geophys. Res. Oceans, 124, 1509-1523, https://doi.org/10.1029/2018JC014645, 2019.
- 670 Vichi, M., Allen, J. I., Masina, S., and Hardman-Mountford, N. J.: The Emergence of Ocean Biogeochemical Provinces: A Quantitative Assessment and a Diagnostic for Model Evaluation, Global Biogeochem. Cv., 25, https://doi.org/10.1029/2010GB003867, 2011.
 - Williams, R. G. and Follows, M.: Ocean Dynamics and the Carbon Cycle: Principles and Mechanisms, Cambridge University Press, 2011.

- Williams, R. G., McDonagh, E., Roussenov, V. M., Torres-Valdes, S., King, B., Sanders, R., and Hansell, D. A.: Nutrient Streams in the North Atlantic: Advective Pathways of Inorganic and Dissolved Organic Nutrients, Global Biogeochem. Cy., 25, n/a–n/a, https://doi.org/10.1029/2010GB003853, 2011.
- Wilson, C.: Global Climatological Relationships between Satellite Biological and Physical Observations and Upper Ocean Properties, J. Geophys. Res., 110, C10001, https://doi.org/10.1029/2004JC002724, 2005.

675

- Wilson, C.: Evidence of Episodic Nitrate Injections in the Oligotrophic North Pacific Associated with Surface Chlorophyll Blooms, J. Geophys. Res. Oceans, 126, https://doi.org/10.1029/2021JC017169, 2021.
- 680 Zhai, X., Greatbatch, R. J., and Kohlmann, J.-D.: On the Seasonal Variability of Eddy Kinetic Energy in the Gulf Stream Region, Geophys. Res. Lett., 35, https://doi.org/10.1029/2008GL036412, 2008.